Author's response to reviews

Title: Detection of events of public health importance under the International Health Regulations: A toolkit to improve reporting of unusual events by frontline healthcare workers

Authors:

Emily MacDonald (emily.macdonald@fhi.no)
Preben Aavitsland (preben.aavitsland@fhi.no)
Dounia Bitar (d.bitar@invs.sante.fr)
Katrine Borgen (katrine.borgen@fhi.no)

Version: 4 Date: 14 September 2011

Author's response to reviews:

Please find attached the latest version of the manuscript. The changes made for this version are limited to some minor typographical and grammatical corrections and are indicated using track changes.